Author's response to reviews

Title: 10 year follow up after Roux-Elmslie-Trillat operation in case of patellar instability

Authors:

Stefan Endres (s.endres@elisabeth-klinik.de)
Axel Wilke (a.wilke@elisabeth-klinik.de)

Version: 2 Date: 22 July 2010

Author's response to reviews:

Dear Ladies and Gentlemen,

enclosed you’ll find the revised manuscript regarding the comments of the editor.

- The name of the ethics committee was added in the method section.
- The competing interest statements can be found at the end of the manuscript.
- Acknowledgment section is not necessary for the present manuscript.

Thanks and best regards

Stefan Endres